Author's response to reviews

Title: Physical Activity during Cancer Treatment (PACT) Study: design of a randomised clinical trial

Authors:

Miranda J. Velthuis (mvelthuis@ikmn.nl)
Anne M. May (a.m.may@umcutrecht.nl)
Ria A.G. Koppejan-Rensenbrink (rkoppejan@ikmn.nl)
Brigitte C.M. Gijsen (b.gijsen@ikl.nl)
Eric van Breda (Eric.vanBreda@BW.unimaas.nl)
G Ardine de Wit (G.A.deWit@umcutrecht.nl)
Carin D. Schröder (c.schroder@dehoogstraat.nl)
Evelyn M. Monninkhof (E.Monninkhof@umcutrecht.nl)
Eline Lindeman (e.lindeman@umcutrecht.nl)
Elsken van der Wall (E.vanderWall@umcutrecht.nl)
Petra H.M. Peeters (P.H.M.Peeters@umcutrecht.nl)

Version: 8 Date: 8 June 2010

Author's response to reviews:

We made some minor revisions in our paper:

1) We replaced 'hypothesized' by 'hypothesised'.
2) Introduction: 1st sentence: we replaced 'maintenaince' by 'maintaining'. We removed the following sentence: 'To date, there is no definitive treatment for fatigue'
3) Methods/design: study endpoints: primary outcomes: We removed one of the descriptions of the subscales (general fatigue....mental fatigue). In the sentence 'Diaries will include ......expenses', we replaced 'such as' by 'also including'.

Hopefully, we provided you with enough information. We are looking forward hearing from you again.

Yours sincerely,

Miranda J. Velthuis
On behalf of all authors
Comprehensive Cancer Center Middle Netherlands
P.O. Box 19079
3501 DB Utrecht
The Netherlands
tel: +31 30 233 7091